Craig Chambers CSE 401 Introduction to Compiler Construction

Readings and Assignment #5

By: Read:

M 4/24 Reminder: Sections 6.5-6.6 M 4/24 MIDTERM REVIEW W 4/26 MIDTERM F 4/28 Sections 7.1, 7.2, 7.3 M 5/1 Section 7.4

By: Do:

M 4/24 Reminder: Complete the implementation of the extended PL/0 parser. F 4/28 Exercises 6.1, 6.3, 6.4, 6.8a, 6.10

M 5/1 Design what additional checks are required to implement the semantic analysis part of the project (a.k.a. the typechecker). Turn in a list of your intended checks.